Author’s response to reviews

Title: Sub-hypnotic dose of propofol as antiemetic prophylaxis attenuates intrathecal morphine-induced postoperative nausea and vomiting, and pruritus in parturient undergoing cesarean section — a randomized control trial.

Authors:

Sylvanus Kampo (sylvanuskampo@yahoo.com)

Alfred Aful (alfredo2k2@yahoo.com)

Shiraj Mohammed (shirajm90@gmail.com)

Michael Ntim (archangelmntim@yahoo.co.uk)

Alexis Buunaaim (abuunaaim@yahoo.co.uk)

Thomas Anabah (dranabah@gmail.com)

Version: 2 Date: 06 Jun 2019

Author’s response to reviews:

Response to editor Comments:

1. We have provided the statistical analysis methods and their results as well as interpretation in our revised manuscript. The test used for each variable has been mentioned. Post-hoc tests were applied. The P values are written for all the categorical variables. Please find our manuscript for your perusal.

2. We have removed duplication of results in the tables and figures.

3. Blinding of randomization is mentioned in the method.

Response to Premal Trivedi, M.D. (Reviewer 1) comments:

1. The exact inclusion and exclusion criteria have been included in our manuscript in the method section.

2. The p-values in the comparisons of hypotension, ephedrine administration, uterine exteriorization, and others have been provided in our manuscript.

3. We did not compare the estimated blood loss (EBL) in this study. However, we determined the EBL for the exclusion criteria (EBL ≥ 500 mL)
Response to Rajagopalan Venkatraman, MD (Reviewer 2) comments:

1. We have described the method used for sample size estimation at the method section in our manuscript.

2. The P values are written for all the categorical variables. Please find our manuscript for your perusal.

3. The P values are included for all the categorical variables.

All other comments raised have been addressed in our manuscript.